# Adopting environmentally-friendly certifications: Transaction cost and capabilities perspectives within the Italian wine supply chain

Stefanella Stranieri

Department of Environmental Science and Policy, Universita degli Studi di Milano, Milano, Italy

Alessandro Varacca and Mirta Casati

Department of Agricultural and Food Economics, Universita Cattolica del Sacro Cuore, Piacenza, Italy

Ettore Capri

Department of Sustainable Food Process, Universita Cattolica del Sacro Cuore, Piacenza, Italy, and

Claudio Soregaroli

Department of Agricultural and Food Economics, Universita Cattolica del Sacro Cuore, Piacenza, Italy

# Abstract

**Purpose** – Environmentally-friendly certifications have increased over the past decade within food supply chains. Although a large body of literature has explored the drivers leading firms to adopt such certifications, it has not closely examined the strategic motivations associated with their adoption. This paper aims to investigate an environmentally-friendly certification, VIVA, examining its role as an alternative form of supply chain governance. The aim is to investigate the drivers affecting the adoption of VIVA and to assess managerial perceptions related to transaction-related characteristics and the firm's internal resources and capabilities.

**Design/methodology/approach** – This study draws upon both an extended transaction cost economics perspective, which is based on transaction risks and the resource-based view, which examines a firm's internal resources. A survey was conducted via a structured questionnaire sent to all of the wine producers in charge of the decision regarding whether to adopt VIVA certification. A Hierarchal Bayesian Model was applied to analyse questionnaire responses. Such a model allows us to specify the probabilistic relationship between questions and latent constructs and to carry over uncertainty across modelling levels.

**Findings** – The adoption of this environmentally-friendly certification is envisioned as a tool to curb internal risks, and thus to manage behavioural uncertainty within the supply chain. A high level of exposure to exogenous transaction risks discourages firms from adopting VIVA certification. The certification system is not perceived as a promoter of operational capabilities. Managers are more likely to implement the certification when they expect that its adoption will leverage their potential knowledge of the supply chain or prompt new and better collaborations with the suppliers. Therefore, the certification can become a resource that interacts with the capabilities of the firm, expressing complementarities that stimulate the formation of dynamic capabilities.

**Research limitations/implications** – The identification of drivers from the two theoretical perspectives offers insights into the attributes that are perceived as important by managers and which, therefore, could be leveraged to foster the adoption of the environmental certification. The external validity of the study could be improved by extending the sample to other certifications and supply chains.

**Originality/value** – The study offers a different perspective on environmental certification. It demonstrates that considering the certification as an alternative form of supply chain governance opens up a set of efficiency and strategic considerations that could be addressed to promote the effectiveness of an environmental strategy within a supply chain

Keywords Governance, Green supply chains, Transaction cost theory, Resource-based view

Paper type Research paper

# **1. Introduction**

Growing concerns about environmental change and resource depletion are increasingly prominent on the world stage. For example, in food supply chains, many environmental concerns

The current issue and full text archive of this journal is available on Emerald Insight at: https://www.emerald.com/insight/1359-8546.htm



Supply Chain Management: An International Journal 27/7 (2022) 33–48 Emerald Publishing Limited [ISSN 1359-8546] [DOI 10.1108/SCM-12-2020-0598] © Stefanella Stranieri, Alessandro Varacca, Mirta Casati, Ettore Capri and Claudio Soregaroli. Published by Emerald Publishing Limited. This article is published under the Creative Commons Attribution (CC BY 4.0) licence. Anyone may reproduce, distribute, translate and create derivative works of this article (for both commercial and non-commercial purposes), subject to full attribution to the original publication and authors. The full terms of this licence maybe seen at http://creativecommons.org/licences/ by/4.0/legalcode

*Financial Disclosure*: This research did not receive any specific grant from funding agencies in the public, commercial or not-for-profit sectors.

Received 2 December 2020 Revised 27 January 2021 5 March 2021 Accepted 8 March 2021

are related to agri-food activities such as the impacts of soil erosion, fertiliser excesses and animal handling practices (Jose and Shanmugam, 2019; Notarnicola *et al.*, 2012). Firms in various industries are increasingly shifting to greener supply chain practices (Stranieri *et al.*, 2019). In food supply chains, a number of green practices have gained ground over the past decade to manage existing environmental challenges and to satisfy increasing consumer demand for environmentallyfriendly products (De Steur *et al.*, 2020).

certifications Voluntary environmentally-friendly are amongst the instruments used by firms to express their commitment to green strategies. To date, there exists a large variety of such instruments (Marette, 2016). They can be both public and private and they can be of an individual or collective nature (Hammoudi et al., 2015). Moreover, they can be either sector-specific or not and they imply the certification of different product and process characteristics (Yokessa and Marette, 2019). Apart from their intent to improve the sustainability of production processes, environmentallyfriendly certifications are considered effective instruments to reduce market failures related to information asymmetry between consumers and producers, to improve stakeholders' reputation on the market and to introduce liability and trust amongst agents within a supply chain. Moreover, these certifications are also used as marketing tools for product differentiation and producers' instruments to reach consumers with a high willingness to pay (Bonroy and Constantatos, 2015).

Current literature has highlighted several categories of drivers influencing the adoption of these certifications: firm-based drivers, institutional drivers and supply chain drivers (Chkanikova and Lehner, 2015; Lee et al., 2013; Naidoo and Gasparatos, 2018). Firm-based drivers refer to an improvement on performance indicators related to cost-effectiveness, the achievement of competitive advantage and a good level of firm-internal knowledge and management involvement regarding environmentally-related topics (Hauschildt and Schulze-Ehlers, 2014; Pullman et al., 2009; Vermeulen and Ras, 2006). Amongst institutional drivers, the pressure exerted by NGOs, civil society, regulatory agencies and consumer preferences has a positive effect on the adoption of green certifications (Ding et al., 2019). Supply chain drivers relate mostly to increased transparency, which is due to an increase in the information exchanged amongst supply chain agents and improved supply chain monitoring activities, which are present because of the enhancement of standardised processes (Dodds et al., 2013; Johnsen et al., 2017; Mangla et al., 2018). Within the supply chain context, environmentally-friendly certifications also act as tools to improve supply chain coordination. More precisely, they can be viewed as alternative forms of organisational governance that lead to an increase in supply chain efficiency due to more transparent transactions and more integrated activities (Ménard, 2017; Stranieri et al., 2017). This improved vertical coordination is related to the introduction of specific agreements amongst supply chain agents for the implementation of such certifications and the centralisation of information flows within supply chains (Passuello et al., 2015).

The decision to adopt an environmentally-friendly certification can be considered both as a solution to the coordination problems related to vertical exchanges (supply chain perspective) and as a solution to improve a firm's competitiveness (firm's *Volume 27 · Number 7 · 2022 · 33–48* 

perspective). From a supply chain perspective, adoption of the certification implies an improvement in transaction transparency and a better distribution of liability amongst supply chain agents. Moreover, the centralisation of supply chain activities leads to a reduction in transaction costs. From a firm's perspective, the certification requires the internalisation of activities and the improvement of vertical relationships, calling for a mobilisation of internal resources and competencies (Hart, 1995; Prajogo *et al.*, 2014; Treacy *et al.*, 2019). This requirement for the development of knowledge within a firm could stimulate dynamic capabilities fostering a sustained competitive advantage.

Despite its importance from a supply chain perspective, there is a limited amount of research on the role of environmentallyfriendly certifications as alternative forms of organisational governance. Managerial motivations and expectations can be very different when choosing certain strategic paths. Understanding the drivers behind decisions relating to these certifications could foster improved design, effectiveness and increased adoption, improving both the management of the supply chain and its environmental impact. Therefore, the present paper aims to investigate the drivers affecting the adoption of these forms of transaction governance in terms of both the supply chain and the firm: i.e. taking into account both the relationships between the certification and transaction-related characteristics and between the certification and a firm's internal resources and capabilities.

To achieve our goal, we drew upon two theoretical frameworks: transaction cost economics (TCE) and the resource-based view (RBV). These two conceptual approaches are chosen as analytical frameworks because they allow us to explore the adoption of environmentally-friendly certifications from both supply chain- and firm-based perspectives. TCE regard the transaction as a basic unit of analysis, underscoring the role of transaction costs in determining efficient supply chain governance (Williamson, 1991). In contrast, RBV regard the firm as the basic unit of analysis (Barney, 2001), focussing on asset integration as a possible way to exploit resources (Argyres and Zenger, 2012) and enhance firm capabilities (Hart, 1995). Taken together, these two theoretical approaches provide a solid theoretical basis for the formulation of hypotheses regarding the strategic drivers behind the adoption of voluntary certification.

This study focusses on the Italian wine supply chain. Specifically, we study the drivers of adoption of the public voluntary certification "VIVA". Data were collected through a survey based on a structured questionnaire sent to all VIVAcertified firms and firms that were considering the implementation of VIVA certification but did not adopt it. We use a Hierarchal Bayesian Model (HMB) to analyse questionnaire responses and shed light on which drivers influence the adoption of the environmental certification.

The present analysis contributes both to theory and practice. From a theoretical perspective, we shed light on how different kinds of transaction uncertainties and related risks can influence the decision to adopt alternative forms of governance such as environmentally-friendly certifications, within supply chains. Results reveal that behavioural uncertainty-related risks influence positively their adoption, while environmentalrelated risks will act negatively on it. Moreover, using RBV, we help to clarify the role of environmentally-friendly certifications as potential resources for a firm. Differently from other studies, we do not look at certification adoption only as the result of the

exploitation of existing capabilities, but we formulate hypotheses testing their role in developing dynamic ones. Results indicate that such certifications are not considered as a strategic resource per se, but as instruments able to leverage a firm's dynamic capabilities related to supply chain coordination skills. From a practical perspective, the theoretical setting allows featuring some key aspects in the design and promotion of an environmentally-friendly certification. For example, TCE arguments highlight how the optimal design of certification is challenged by the trade-off between efficient coordination of the supply chain and the need for flexibility a firm has when perceiving high external risks. Differently, hypotheses from RBV highlight how a long-run environmental strategy could be fostered by a certification able to stimulate sustainable competitive advantage throughout potential complementarities between the implemented standardised procedures and firm internal resources.

The paper is organised as follows. Section 2 presents the theoretical frameworks, proposing a number of hypotheses formulated based on the TCE and RBV frameworks. Section 3 describes the VIVA certification, the study data, the measures and the empirical model. Section 4 presents the results, which are discussed in Section 5. Finally, Section 6 concludes the paper and outlines limitations and suggestions for future research.

## 2. Theoretical framework and hypotheses

2.1 Transaction cost economics and transaction risks Following TCE, high transaction costs lead to the need for adopting transaction governance forms with a high level of vertical coordination. Higher transaction costs result from the variation of transaction-related characteristics, especially the degree of asset specificity and the degree of transaction uncertainty (Williamson, 1985). The positive variation of one or both such attributes leads to the risk that economic agents act opportunistically. The present study makes large use of TCE theory. Specifically, we refer to an extended perspective that addresses the role of transaction risks in shaping the organisation of supply chains by considering, not only the risk of opportunistic behaviour but also the risks surrounding the transaction which do not hinge on economic agents' behaviour (Stranieri et al., 2017; Wever et al., 2012). Indeed, besides transaction-related risks during vertical exchanges, there are additional types of risks that do not influence transaction characteristics directly but can affect the success of transaction execution (Kaplan and Mikes, 2012). Consequently, forms of transaction governance can be viewed as tools that not only minimise transaction costs but also manage all the risks surrounding vertical exchanges (Wever et al., 2012).

According to Williamson (1985), transaction uncertainty depends on both behavioural uncertainty and environmental uncertainty. The higher the level of behavioural uncertainty, the higher the risk of a transaction failure due to noncompliance with the conditions set up within agreements (Williamson, 1991). This kind of transaction risk is defined as "internal" because it can be managed within the transaction, through the adoption of a form of governance that reduces behavioural uncertainty through improved transaction transparency. In general, the higher a transaction's internal Volume 27 · Number 7 · 2022 · 33–48

risks, the greater the probability that a firm will adopt forms of transaction governance with a high level of vertical coordination (Fernández-Olmos et al., 2009; Stranieri et al., 2017). As noted above, environmentally-friendly certifications can be considered alternative organisational governance forms that improve transaction transparency and foster coordination amongst supply chain partners. Indeed, these instruments imply the adoption of agreements that set out specific activities and rules that supply chain agents need to follow. Such agreements can establish standard procedures, guidelines for the operation of certified activities and the distribution of specific responsibilities to economic agents. Information flows are, thus, centralised, implying that the firm receiving the certification acts as the leader of the supply chain and stimulates its partner to implement and monitor the standards. Such a firm usually coordinates the implementation of certified activities within the supply chain, also internalising monitoring activities related to the correct execution of certified procedures throughout the supply chain.

According to the transaction cost perspective, the adoption of such certifications may be linked to the presence of internal risks surrounding transactions within the supply chain. Thus, we hypothesise:

*TCE-H1.* The higher the firm's perception of transactioninternal risks, the higher the probability that it will adopt an environmentally-friendly certification.

Environmental uncertainty is not caused by the behaviour of supply chain agents in the accomplishment of vertical exchanges. Instead, it depends on unpredictable changes in the economic environment that may lead to an increased risk of maladaptation amongst supply chain agents (Williamson, 1985). Environmental uncertainty and transaction-related risks lead to increased difficulty in the arrangement of agreements (Gurcaylilar-Yenidogan and Windsperger, 2014). Such risks are defined as "transaction-external" risks, as they are exogenous to the behaviour of transacting parties and have a strong influence on the management of conditions set up in the agreements (Sydow et al., 2013; Wever et al., 2012). Different types of transaction-external risks have been found to influence transaction execution within food supply chains. Variations in agricultural raw material and product prices, the complexity related to frequent changes in legislative frameworks and the changing quality preferences of consumers affect the ease of accomplishing transaction conditions and achieving transaction efficiency. For example, Hau (2006) stresses the positive relationship between transaction costs and financial volatility, while Huchet-Bourdon (2011) analyses agricultural commodity price volatility, highlighting that high price events have characterised agri-food system exchange dynamics for the past 50 years. Moreover, Farber et al. (2011), Kvalvik et al. (2011) and Grimm et al. (2014) discuss how changing agricultural policies on food safety and climate change play an important role in a firm's adaptive capacity and on sustainable supply chain relationships. Beske et al. (2014) discuss the challenges that food supply agents face in adapting to constant changes in consumer preferences.

The debate surrounding the influence of external risks on the governance of vertical relationships is still inconclusive, especially concerning hybrid (intermediate) forms of transaction governance (Ménard, 2017; Romme, 1990; Sydow *et al.*, 2013). Transaction

Italian wine supply chain

#### Stefanella Stranieri et al.

cost theory assumes that supply chain agents will adopt hierarchical forms of transaction governance when external transaction risk is high (Williamson, 1985). However, current empirical evidence highlights a negative relationship between the level of transaction-external risks and the level of vertical coordination (Das and Teng, 2001; Olmos, 2010). Regarding public certifications, for example, the collective nature of such instruments means that suppliers are substitutable with competitors. Therefore, external transaction risks could lead to a higher risk that certified firms will be substituted for other certified firms in cases of delayed adaptation to the changing economic environment (Hartmann et al., 2010). In this situation, the transaction costs deriving from external risks cannot be mitigated by tighter coordination throughout the adoption of the certification (Schulze et al., 2007). For these reasons, supply chain agents could decide to remain flexible and adapt to environmental changes through more flexible forms of transaction coordination (Perrow, 1986).

Based on the above discussion regarding the role of transaction-external risks in affecting vertical coordination and the role of different public certifications within food supply chains, we propose the following hypothesis:

*TCE-H2.* The higher the firm's perception of transactionexternal risks, the lower the probability that it will adopt an environmentally-friendly certification.

#### 2.2 The resource-based view

In its initial stages, the RBV considered a firm as a unique bundle of resources that, if adequately used, bring sustained competitive advantage and greater value to the firm (Barney, 1986; Wernerfelt, 1984). In subsequent literature, this view was extended to distinguish resources and capabilities: resources include the tangible and intangible assets available to a firm for the production process such as equipment, skills, processes or finance (Barney, 2001; Grant, 1991); capabilities refer to the capacity to deploy a subset of resources to perform an activity with the desired goal (Amit and Schoemaker, 1993).

Resources are assumed to be heterogeneous and not perfectly transferable across firms (Barney, 1986; Wernerfelt, 1984). When these resources are valuable, rare, inimitable and non-substitutable (VRIN) (Wernerfelt, 1989), they can give rise to competitive advantage. Firms tend to integrate those assets and activities that have a unique complementarity with existing ones (Argyres and Zenger, 2012). According to the RBV, assets should be integrated when bundled with those already present within the firm, they bring unique and higher added value as compared to other firms. An environmentallyfriendly certification is essentially an asset that, if integrated into the firm, plays the role of an organisational resource that coordinates various activities within the firm and the supply chain. Consequently, it can be considered a form of governance that improves vertical coordination. Such a certification is collective in nature, as it includes replicable practices by all firms adopting it (Treacy et al., 2019). However, it can also possess characteristics that are unique to a firm, depending on its complementarities when bundled with existing assets. This assessment pertains to the views of the company's managers. Therefore, the decision to adopt (integrate) the certification *Volume 27 · Number 7 · 2022 · 33–48* 

can depend on their perception of the VRIN characteristics. Hence, we can propose the following hypothesis:

*RBV-H3.* The higher the firm's perception that an environmentally-friendly certification has VRIN characteristics, the higher the probability of its adoption.

Competitive advantage depends not only on resources but also on the way that they can be leveraged through a firm's capabilities. Capabilities can be considered a distinctive source of competitive advantage, as they cannot easily be copied, unlike the underlying resources (Collis and Montgomery, 1995). Several studies in the review by Argyres and Zenger (2012) postulate that firms tend to integrate an asset when they have higher prior capabilities for performing related existing activities than upstream or downstream actors. These higher capabilities predict the better performance of the firm in those specific activities. Environmental certifications can improve the efficiency of internal operations because of the scrutiny and constant monitoring of processes. This can be associated with an increase in internal operational capabilities (Darnall and Edwards, 2006). Additionally, the training of employees necessary for the implementation of certification and the resulting enhancement of interpersonal relationships can increase operational capabilities and improve the production process and labour productivity (Delmas and Pekovic, 2013; Lo et al., 2014). Better internal coordination practices and control of costs also lead to higher efficiency in production planning, with lower production costs (Treacy et al., 2019) and a better focus on value-adding production processes (Lo et al., 2009). Supply chain coordination is also involved: the adoption and diffusion of an environmentallyfriendly certification within a company can improve the procurement of inputs, as well as the adoption of green practices along the supply chain (Prajogo et al., 2014). Finally, focussing on wine, Annunziata et al. (2018) show a positive relationship between proactive socio-environmental practices and cost advantages related to the operational dimension of the firm. Therefore, managers can expect increased efficiency due to the development of higher operational capabilities. Based on the above, the following hypothesis can be formulated:

*RBV-H4.* The higher the firm's perception that an environmentally-friendly certification promotes the coordination of operational capability, the higher the probability of its adoption.

To produce a sustained competitive advantage, integration choices also need to reflect market dynamism. In other words, they need to reflect complementarities with the supply chain environment (Nagano, 2020). When higher market dynamism is present, knowledge resources become important, as they can spur innovation and adaptation to new market conditions. Dynamic capabilities – the ability to manipulate firm and supply chain resources and gain competitive advantage – become central. This manipulation, which also includes entrepreneurship (Teece, 2009), allows for the development of new value-creating strategies (Santos and Eisenhardt, 2005). Capabilities only bring sustained competitive advantage when they are dynamic in nature as they "emphasise a firm's constant pursuit of the renewal, reconfiguration and re-creation of resources" (Wang and Ahmed, 2007, p. 36).

Gavronski *et al.* (2011) show that the adoption of certain routines such as those linked to green manufacturing processes and ISO 14001, provides the basis for the formation of new and dynamic capabilities. Within the wine sector, Annunziata *et al.* (2018) show that two types of organisational capabilities are associated with the adoption of proactive socio-environmental practices: the capacity for collaboration with partners and suppliers and the capacity for product innovation.

Collaboration with suppliers and more generally with stakeholders is suggested in the literature as one of the necessary prior capabilities favouring the adoption of environmental management practices (Hofmann *et al.*, 2012). At the same time, an environmental certification can promote the development of these capabilities because of the integration of practices with supply chain partners and suppliers. Under the coordinating role of the certification, firms can gain better knowledge about products, suppliers and supply chain characteristics and can start new partnerships exploiting their potential knowledge (Ghozzi *et al.*, 2016). This, in turn, could improve the firm's capacity to coordinate supply chain and stakeholders, fostering a long-run environmental strategy (Hart, 1995). Therefore, assuming market dynamism, we can formulate the following hypothesis:

*RBV-H5.* The higher the firm's perception that an environmentally-friendly certification promotes the development of a firm's supply chain coordination capacity, the higher the probability of its adoption.

As stated by Annunziata et al. (2018), the product innovation process leads to the development of internal knowledge, the acquisition of external knowledge and a general learning process within the firm (Cassiman and Veugelers, 2006; Jiménez-Jiménez and Sanz-Valle, 2011). When innovation is of a sustainable nature, it "increases organisational maturity and provides strategic and economic viable novelty and competencies that can foster actions to address the challenge of corporate sustainability" (Annunziata et al., 2018, p. 1302). An environmentally-friendly certification can favour the innovation process, allowing access to new and more advanced procedures and improved introduction and tracking of new practices, which, in turn, stimulate the introduction of new ideas for the production processes. To have this effect, the certification needs to be embedded in daily routines and leverage capabilities across various functions of the firm. For example, Prajogo et al. (2014) show that with greater internalisation (diffusion) of a green management system such as certification, within the different functions of a firm, more green product practices (such as new product designs) are adopted. Hence, still assuming market dynamism, we can propose that:

*RBV-H6.* The higher the firm's perception that an environmentally-friendly certification promotes the development of innovation capacity, the higher the probability of its adoption.

# 3. Data and empirical model

## 3.1 The VIVA certification

As introduced above, the present analysis refers to VIVA, a sustainability-related certification used in the Italian wine sector. The motivations related to this choice are three-fold.

## Volume 27 · Number 7 · 2022 · 33–48

Firstly, the wine context is particularly interesting as it currently confronts multiple threats from the natural world such as water scarcity, climate change and volatile energy prices (Gilinsky *et al.*, 2016). Moreover, wine is one of the main crops for most of the terrestrial agroecosystems in Europe and in the temperate climate zones worldwide. Increased consumer interest in sustainable viticulture and the potentials for technical improvements in vineyards have led wine companies to take a greener approach by adopting different voluntary codes (De Steur *et al.*, 2020; Jourjon *et al.*, 2016).

Secondly, Italy has consistently been a worldwide wine production leader. In 2018, wine performed best across national agricultural products, with a production value of  $\notin$ 10.2bn [Italian National Institute of Statistics (ISTAT), 2019]. According to the latest publicly available data, in 2014 over 500 Italian wineries were directly involved in sustainable development networks. Those companies accounted for at least one-third of the total wine production value, with an estimated turnover of about  $\notin$ 3.1bn [Forum per la sostenibilità del vino (FSV), 2014].

Thirdly, although a number of environmentally related certifications have been proposed in the Italian territory (Borra et al., 2016), in 2020 only five programmes are active. Of these, VIVA is the only certification promoted by a public institution, the Italian Ministry for the Environment, Land and Sea [1]. VIVA started as a pilot project in 2010 and within 10 years had successfully built a network of 79 certified firms, with 64 different types of wine bottles certified. We choose to focus on such certification as it refers to specific sustainability indicators that are directly measurable and not only declared. This allows concentrating on firms that fully embraced the certification in their sustainability strategy. The performance analysis to obtain the certification is conducted using four indicators: air, water, vineyard and territory. The first three indicators are more related to agronomic management, thus strongly emphasising the environmental facet of sustainability. The fourth indicator embraces the social sustainability aspect, linking wine to its territory and community. The project is open not only to wineries and processing firms but also to bottlers and a company can choose whether to certify the whole organisation, one product or several products. As a final characteristic, VIVA includes the issuance of a label and a QR code on the bottle. The label increases transparency and makes the certification visible to the consumer, thus making the environmentallyfriendly attribute clearer [2].

## 3.2 Data

To test the six hypotheses proposed in the previous section, we collected data through questionnaires consisting of 21 statements evaluated on a seven-point Likert scale (where 1 indicated very strong disagreement and 7 represented very strong agreement). Each statement addressed one of six latent characteristics that might lead wine companies to implement (or not implement) the VIVA certification. These unobservable features relate directly to the previously discussed theoretical predictions. For example, hypotheses *TCE-H1* and *TCE-H2* can be reformulated as functional relationships between firms' exposure to internal and external risk, respectively, and the probability of adopting the certification. As there is no way to directly quantify the degree of internal and external risk

incurred by companies, these measures are said to be latent. Then, from a regression perspective, our problem consists of studying the expected change in the (log-) odds of getting the certification when comparing firms differing by one unit of the hidden attribute. The same conceptual framework applies to hypotheses *RBV-H3* to *RBV-H6*, only this time the unobserved features of interest concern the perceived benefits (or lack thereof) of implementing the environmentally-friendly certification. Table 1 briefly introduces the relationship between our 21 statements and the six latent characteristics they aim to approximate. In the next section, we discuss how we extracted latent characteristics from Likert-valued statements through a simple Bayesian multilevel model.

To improve internal validity, data collection was conducted using a two-stage approach where, for each respondent, an interview was performed before the survey. In the interview phase, we first contacted the companies by email explaining the purpose of the study and anticipating our phone call. The questionnaires were attached to the email for a preview. The initial submission involved 148 Italian wine producers that had recently shown interest in the VIVA certification. The list of such firms was provided by the Catholic University *Volume 27 · Number 7 · 2022 · 33–48* 

Sustainability Research Centre, which was directly involved by the Italian Ministry for the Environment in the creation and development of the certification. These companies had contacted the Centre to investigate and obtain more detailed information about the VIVA certification. Of these firms, 46 had already implemented the VIVA certification at the time of the interview (Group A), while 102 had not implemented the certification at the time of the interview (Group B).

Then, we contacted the contact persons via telephone. The purpose of the interview was to assure the data quality. It allowed establishing whether:

- the interviewees matched our preferred employee profile;
- the firms in Group B had seriously considered the VIVA certification or if it was just a spurious contact; and,
- the respondent was willing to complete the survey.

To ensure data trustworthiness, we decided to interview only those actors who were directly involved in the decision-making process regarding VIVA certification within the firm, i.e. firm owners, quality assurance managers and technical directors. If all these three conditions were met, we proceeded briefly to illustrate the scope and the content of the questionnaire. The interview closed with an appointment for a *vis-à-vis* meeting or,

 Table 1
 Structure of questionnaires and overview of responses

Latent characteristics	Statements	<b>≤3</b>	>4	
<i>TCE-related variables</i> Internal risk	All of the transactions with suppliers ended unsuccessfully	18	74	
(TCE-H1)	The conditions laid down with suppliers were not respected It is not possible to predict in advance the exact quantity of raw material sourcing	13 21	79 71	
External risk	Variations in agricultural raw material prices are a risk for the company	33	59	
(TCE-H2)	The complexities embedded in the legislative framework are a risk for the company	26	66	
	Qualitative variations in consumer's preferences are a risk for the company	30	62	
	Variations in wine prices could be a risk for the company	29	63	
RBV-related variables VRIN	The environmentally-friendly certification can be a source of competitive advantage	9	83	
characteristics	My competitors have not already implemented an environmentally-friendly certification	52	40	
of the resource	It is hard to implement an environmentally-friendly certification	61	31	
(RBV-H3)	It is hard to switch from one certification system to another	48	44	
Operational capabilities	Through the implementation of the certification, my firm can gain more certainty concerning costs along the chain	33	59	
(RBV-H4)	The environmentally-friendly certification can be a tool to optimise production and logistics processes The environmentally-friendly certification could allow my company to focus on its most important activities	27 20	65 72	
Dynamic capabilities related to the supply chain	I think that new competencies can emerge exploiting internal or external synergies created through the certification implementation	12	80	
(RBV-H5)	I think that the environmentally-friendly certification could allow my company to broaden its knowledge of product characteristics, the supply chain and suppliers	11	81	
	I think that the environmentally-friendly certification could offer the possibility of adopting new commercial relationships along the supply chain that could be hard to engage in other ways	19	73	
Dynamic capabilities related	The environmentally-friendly certification can be a stimulus for generating new ideas for production processes	7	85	
to innovation (RBV-H6)	I think that the environmentally-friendly certification could allow my company to access new and more advanced procedures	9	83	
	I think that the environmentally-friendly certification could allow my company to keep track of new practices	9	83	
	I think that the environmentally-friendly certification could allow my company to introduce new practices	9	83	
Note: *Number of answers pro	oviding scores $\leq$ 3 or $>$ 4 in the 1–7 Likert scale			

when not possible, for another telephone call where we could move to the survey phase.

Administering the survey vis- $\dot{a}$ -vis or by phone allowed to ascertain that the replies coherently reflected the goals of the questionnaire. Further, compiling the survey in this form allowed us to elicit the degree of understanding of the topics discussed, thus providing the opportunity to clarify misunderstandings about any specific statement or measurement. The latter is a key aspect for the reliability (Yin, 2017) of our research: as assessing 21 statements can be challenging and, as some concepts might not be immediately clear to the respondent, active guidance cannot be easily separated from the survey.

Amongst the 46 certified firms, 36 agreed to participate in the survey (78% response rate), while another 56 responses came from the group of 102 uncertified wine producers (55% response rate). On average, interviewees had been working in the wine industry for roughly 17 years and had been used by their current company for nearly 13 years. On average, most respondents had been in the same job position for 15 years.

The questionnaires administered to groups A and B shared the same structure and content, except that the interviewer was instructed to read Group A one additional introductory sentence. That is, the Group A survey started with the phrase: "When you decided to adopt the sustainability certification [...]." and progressed by clarifying that replies should reflect the respondents' opinions at the time the certification was implemented. The Group B questionnaire did not have an introductory sentence, so wine producers were asked to express their beliefs in terms of the company structure at the time of the interview.

Table 2. Sample characteris	stics <sup>a</sup>
-----------------------------	--------------------

Volume 27 · Number 7 · 2022 · 33–48

As outlined in Table 2, nearly 90% of the interviewees are located in the north-eastern part of Italy (e.g. Trentino-Alto Adige/Sudtirol, Veneto, Friuli-Venezia Giulia and Emilia Romagna). In total, 42% of the firms market their products through the Ho.Re.Ca. distribution channel, while 33% sell through large-scale retailers. Small producers are almost equally represented, leaving medium producers as approximately 26% of the observations. The production level is not evenly distributed amongst certified and non-certified firms. Table 2 shows that 56% of the former produce less than 500,000 bottles per year, whereas 46% of the latter market over 1,000,000 bottles per year. Furthermore, our data set indicates that while most small and medium producers sell their product through the Ho.Re.Ca. the channel, large enterprises typically merchandise wine bottles via large-scale distributors. Finally, 80% of the companies interviewed already had a sustainability strategy in place, while 43% of them had already implemented other environmentally-oriented certifications (e.g. ISO14001, carbon footprint labels and/or organic certifications).

## 3.3 Empirical model

Given the six latent characteristics presented in Table 1, we need a probabilistic model linking these unobservable firm attributes to specific sets of Likert-valued inquiries designed to elicit the corresponding underlying feature. In this context, the advantage of using HMBs is twofold: firstly, one can fully specify the probabilistic relationship between questions and latent constructs using any suitable theoretical framework; secondly, HBMs allow for the carrying over of uncertainty

Characteristics	Group A (certified) (%)	Group B (not certified) (%)	Total (%)
Facility location <sup>b</sup>			
Northeast Italy	36	39	38
Northwest Italy	11	14	13
Central Italy	28	27	27
South Italy	11	13	12
Insular Italy	14	7	10
Production volume <sup>c</sup>			
Low	14	46	34
Medium	31	23	26
High	56	30	40
Distribution channel			
Ho.Re.Ca.	31	50	42
Large-scale retail	47	23	33
E-commerce	8	9	9
Wholesaler	11	11	10
Other	3	7	7
Existence of firm sustainability strategy			
No	17	21	20
Yes	83	79	80
Other environmental certifications/practices within firms			
No	34	59	57
Yes	47	41	43
<b>Notes:</b> ${}^{a}n = 92$ ; ${}^{b}NUTS1$ level in EU classification; ${}^{c}Low = <500, C$	000; medium = 500,000 < bottles <	1,000,000, high = >1,000,000	

across modelling levels, thus removing the need to correct nuisance parameters along the hierarchy.

The first part of our modelling approach consists of arranging individual respondents according to the hidden attributes discussed, thus, far. This empirical sorting is conditional on the rating scores obtained from the polytomous data collected through the surveys. Moreover, as scoring ordered data assuming fixed intervals between the categories can often be unrealistic, we let such distances adjust flexibly. When a latent characteristic is measured through multiple inquiries in a suitably structured questionnaire, item response theory (IRT) models provide a solid theoretical underpinning to link individual responses to unobserved features of the respondents. In particular, when participants are asked to respond to Likert-valued statements (assuming that the lowest/ highest values always indicate strong disagreement/agreement), the rating scale model (RSM-Andrich, 1978) represents the simplest construct relating response scores to item-specific (i.e. statementspecific) and person-specific (latent) components, which are called "item difficulty" and "person-ability", respectively. In contrast to other IRT models such as the partial credit model (PCM - Rasch, 1961), the RSM constrains the distance between item difficulty values to be the same across all items. Clearly, this assumption is only reasonable when item responses are elicited using a fixed set of behavioural thresholds (i.e. Likert-type thresholds). In such cases, the RSM provides a more parsimonious alternative to the PCM (Andersen, 1997). Moreover, as all parameters in the RSM represent locations on an underlying variable, they specifically enable objective comparisons of persons and items (Rasch, 1977).

Mathematically, the RSM model can be formulated using the following category response function (Engelhard, 2014):

$$\mathbf{P}_{ijx} = \frac{\exp\left[\sum_{m=0}^{s} \left(\theta_{j} - \alpha_{i} - \delta_{m}\right)\right]}{\sum_{r=0}^{s} \exp\left[\sum_{m=0}^{r} \left(\theta_{j} - \alpha_{i} - \delta_{m}\right)\right]}$$
(1)

where  $i \in \{1, ..., N\}$  indicates the ith item,  $j \in \{1, ..., M\}$  represents the jth person,  $x \in \{1, 2, ..., S\}$  is the response given by person j to any item i,  $P_{ijx}$  indicates the probability that person j answers x to item i,  $\alpha_i$  stands for the ith item's difficulty and  $\theta_j$  refers to the jth person's ability. We will hereafter re-express equation (1) more compactly as:

$$\mathbf{x}_{ij}|\boldsymbol{\theta}_{j}, \boldsymbol{\alpha}_{i}, \boldsymbol{\delta} \sim \text{RSM}(\boldsymbol{\theta}_{j}, \boldsymbol{\alpha}_{i}, \boldsymbol{\delta})$$
 (2)

where  $\delta$  is an S - 1 vector of thresholds.

Our implementation of the RSM is, however, conceptually different from its standard applications in psychrometric analysis. In fact, we are not interested in modelling respondents' abilities, nor do we want to explicitly correct for item difficulty. Rather, we borrow the above probabilistic constructs to map sets of Likert-valued statements onto continuous measures defined on a common support. Therefore, to make IRT terminology more suitable to our analytical framework, we will henceforth refer to the subscripts i, j as "statement" and "firm", respectively, instead of "item" and "person". Additionally, we will treat the ability parameters,  $\theta$ , as firms' latent characteristics, while  $\alpha$  will hereafter indicate statement-related random bias terms. Although most works focus on the estimation of  $\alpha$  and  $\delta$  to represent P<sub>ijx</sub> as a function of  $\theta$ , we are interested in estimating the latent characteristics themselves [3]. As discussed in

*Volume 27 · Number 7 · 2022 · 33–48* 

Section 3, the TCE and RBV frameworks suggest that firms may choose the VIVA certification depending on six such unobservable attributes [4], thus requiring the extension of equation (2) to:

$$\mathbf{x}_{cij}|\boldsymbol{\theta}_{cj}, \boldsymbol{\alpha}_{ci}, \boldsymbol{\delta} \sim \text{RSM}(\boldsymbol{\theta}_{cj}, \boldsymbol{\alpha}_{ci}, \boldsymbol{\delta})$$
 (3)

where c = 1, ..., 6 works as a placeholder for the aforementioned unobserved features. Moreover, the above statement also implies:

$$\operatorname{Cert}_{j|\pi j} \sim \operatorname{Bernoulli}(\pi_{j})$$
$$\operatorname{logit}(\pi_{j}) = \mu + \theta_{j}' \gamma + \mathbf{z}_{j}' \beta \tag{4}$$

where Cert<sub>j</sub> is a dummy variable taking on the value 1 when firm j adopts the certification,  $\boldsymbol{\theta}_{j} = (\theta_{1j}, \dots \theta_{6j})$ ,  $\mathbf{z}_{j}$  indicates a D-vector of controls and  $\gamma$ ,  $\boldsymbol{\beta}$ , respectively represent  $6 \times 1$  and D  $\times 1$  vectors of fixed effects. The variables in  $\mathbf{z}_{j}$  include a set of binary controls accounting for:

- the presence of other sustainability certifications;
- the retail channel (wholesalers, HoReCa or e-commerce) used to merchandise the product; and
- how costly it was to implement the VIVA standards (we classify these costs as either "high" or "low").

The last element in  $z_j$  is continuous regressor capturing the total production volume, expressed in bottles per year.

Bayesian hierarchical modelling allows the estimation of parameters in equations (3) and (4) jointly, thus propagating the uncertainty in first-stage calculations to the second-stage regression. Except for trivial exercises with canonical and conjugate distributions, this framework requires two basic ingredients:

- 1 the full posterior distribution of the relevant coefficients; and
- 2 a sampling algorithm that efficiently explores the typical set of such a probability function (Betancourt and Girolami, 2015; Gelman *et al.*, 2013; Hoff, 2009; Kruschke, 2014).

These samples are then used to construct summary statistics and posterior credible intervals for the (random) quantities of interest. Besides equations (3) and (4) – where the latter indicates the likelihood function – defining marginal posterior densities for  $\theta$ ,  $\alpha$ ,  $\delta$ ,  $\gamma$  and  $\beta$  require a full array of priors/ hyperpriors distributions. In this work, we set [5]:

$$\begin{aligned} \alpha_{\rm c}|_{\sigma{\rm c},\alpha} &\sim {\rm MVN}(0, {\bf I}_{\rm N_c} \ \sigma_{{\rm c},\alpha}) \\ \sigma_{{\rm c},\alpha} &\sim {\rm Half} - {\rm Normal}(0,3) \\ \theta_{\rm c}|\sigma_{{\rm c},\theta} &\sim {\rm MVN}(0, {\bf I}_{\rm M} \ \sigma_{{\rm c},\theta}) \\ \sigma_{{\rm c},\theta} &= 0.5 \\ \delta &\sim {\rm MVN}(0, \ {\bf I}_{{\rm s}-1} \ \sigma \ \delta) \\ \sigma &\delta &= 3 \end{aligned}$$
(5a)

for the RSM and

$$\mu \sim \text{Normal}(0, 1.5)$$
  
 $\gamma_{c} \sim \text{Normal}(0, 2.5)$   
 $\beta_{d} \sim \text{Normal}(0, 2.5)$  (5b)

for the logistic component (here  $d \in \{1, ..., D\}$ ). In equation (5a),  $\alpha_c = (\alpha_{c1}, ..., \alpha_{cN_c})$ ,  $\theta_c = (\theta_{c1}, ..., \theta_{cM})$ , while  $N_c$  indicates the number of items adapted to measure latent characteristic c.

The choice of modelling  $\alpha$  and  $\theta$  through normal distributions is standard in the literature on random-effects models and other Bayesian implementations of IRT (Bürkner, 2019; Gelman et al., 2013; Luo and Jiao, 2018). Consistent with the growing body of literature discussing the shortcomings of using flat uninformative priors (Gelman et al., 2008; Gelman et al., 2017; Lemoine, 2019; McElreath, 2020), we give  $\sigma_{c,\alpha}$  in equation (5a) a zero-centred, positive-valued weakly informative symmetric distribution [6] as suggested by Bürkner (2019), Luo and Jiao (2018) and Stan's reference manual (Stan Development Team, 2018). On the other hand, following the default distributional choices indicated by Furr (2017) and Gelman et al. (2008), we set the remaining hyperparameters  $\sigma_{c,\theta}$  and  $\sigma_{\delta}$  to 0.5 and 3, respectively [7]. By reflecting the scale of the data entering equation (4),  $\sigma_{c,\theta} = 0.5$ allows us to properly set the corresponding regression parameters. In this respect, Gelman et al. (2008) present some technical guidelines for setting sensible weak priors on fixed effects in logistic regression. Provided that all variables in equation (4) have a mean of 0 and a standard deviation of 0.5 [8], these authors show that a 2.5-scaled Cauchy distribution represents a reasonable default choice for  $\gamma$  and  $\beta$ . However, recent contributions have shown that this setup can be rather informative when - as in our application - information in the data is limited (Ghosh et al., 2018). Therefore, common practice has now shifted to adopting t distributions with 3 or more degrees of freedom or zero-centred normals with standard deviations of up

*Volume 27 · Number 7 · 2022 · 33–48* 

to 2.5 (our choice for the current work). Finally,  $\mu$  indicates the average proportion of Cert when all other variables are set to zero (which, in this case, coincides with their mean). Because this coefficient enters an inverse-logit function, priors defined over a parameter range beyond 4 to -4 would produce posterior predictions either at 1 or 0. Consequently, any prior with a large probability mass beyond these two cut-offs would produce undesirable bimodal posteriors. Therefore, a sensible solution is to adopt a normal distribution with zero location and a 1.5 standard deviation. The inverse-logistic image of this parametrisation produces an approximately uniform density over the 0 – 1 interval, as desired. Further discussion of how modern Bayesian approaches deal with the choice of prior distributions is available in Gelman *et al.* (2017) and McElreath (2020).

## 4. Results

The joint posterior distribution obtained by combining equations (3), (4), (5a) and (5b) can be directly sampled using modern Markov Chain Monte Carlo (MCMC) algorithms [9].

One interesting feature of Bayesian estimation lies in the possibility of complementing the standard information provided by (first/second) moments and confidence intervals with other useful quantities such as posterior quantiles and probabilities. Table 3 reports the estimated posterior medians, 90% and 95% credible intervals (hereafter CIs) and probabilities P(y > 0) and P(y < 0), where † indicates any random variable of interest.

Our estimates support the first hypothesis (*TCE-H1*) presented in Section 2, as highlighted by the positively skewed CIs and the prominent proportion of positive  $\gamma_1$  coefficients. Although the former also includes some negative values, the above-zero probability mass approaches 0.9. This indicates a low uncertainty when discussing the direction of the effect. However, the CI is very large, implying that the magnitude of  $\gamma_1$  remains volatile. Moving to the effect of environmental uncertainty, hypothesis *TCE-H2* is strongly supported. As expected, we find that higher external risk reduces the odds of choosing to certify. The entire posterior distribution of the  $\gamma_2$ 

Variable	Median 95% C.I.		90% C.I.		P(y>0)	P(y<0)	
μ – intercept	-1.02	-2.06	-0.03	-1.89	-0.20	0.02	0.98
		Latent Cha	aracteristics				
γ <sub>1</sub> – TCE: Internal risk	1.419	-0.93	4.08	-0.51	3.60	0.890	0.110
$\gamma_2$ – TCE: External risk	-4.60	-7.51	-2.43	-6.96	-2.74	0	1
$\gamma_3 - RBV: VRIN$	-0.36	-3.16	2.24	-2.62	1.85	0.40	0.60
$\gamma_4$ – RBV: Operational capabilities	0.49	-2.06	3.23	-1.62	2.68	0.66	0.34
$\gamma_5$ – RBV: Capabilities – supply chain	2.54	0.38	5.45	0.71	4.86	0.99	0.01
$\gamma_6$ – RBV: Capabilities – innovation	1.10	-2.00	4.38	-1.44	3.81	0.76	0.24
		Control	variables				
β <sub>1</sub> – Ho. Re. Ca.	0.54	-1.45	2.42	-1.13	2.10	0.71	0.29
$\beta_2$ – e-commerce	1.76	-1.28	4.86	-0.80	4.36	0.87	0.13
$\beta_3$ – Large-scale retailer	2.87	0.87	5.17	1.19	4.83	1	0
$\beta_4$ – Other certifications	0.11	-1.65	1.92	-1.35	1.60	0.56	0.44
$\beta_5$ – Production volume	0.80	-1.01	3.25	-0.74	2.78	0.80	0.20
$\beta_6$ – Certification costs	-2.09	-3.91	-0.46	-3.61	-0.69	0.01	0.99

parameter extends to the left of zero, indicating a rather strong consistency of the negative sign.

Turning to the next block of covariates, the VRIN latent regressor shows the centrality of the parameter, as highlighted by the roughly symmetric share of positive and negative parameter values. This means that the hypothesis *RBV-H3* is not supported by the results. Similarly, hypothesis *RBV-H4*, concerning operational capabilities, is also not supported. Differences in the probability of certifying are essentially undefined, given the relatively symmetric intervals that are centred around zero.

Results become much more clear-cut when the VIVA certification is envisioned as a tool to enhance dynamic capabilities related to the supply chain. In fact, hypothesis *RBV-H5* is strongly supported, as indicated by the positive CIs and by the 99% share of above-zero values for the corresponding  $\gamma$  parameter. On the other hand, the parameters' estimates are less conclusive when referring to dynamic capabilities related to innovation. In fact, for hypothesis *RBV-H6*, 76% of the sampled  $\gamma_5$  parameters lie above zero and the CIs include both positive and negative values. Therefore, although the median value would suggest a positive relationship between trusting the certification's effectiveness in fostering innovation and the odds of certifying, this conclusion remains highly uncertain.

Finally, the control variables indicate that high certification costs decrease the odds that a firm will become certified, while the opposite is true for large-scale marketing channels (compared to wholesalers). Little can be concluded about the extent and direction of the marginal effect for the e-commerce/ Ho.Re.Ca. distribution channels or for the presence of other green certifications. Aside from e-commerce, for which largely positive CIs support the positive location parameter,  $\beta_1$ ,  $\beta_4$  and  $\beta_5$  have positive medians, but their posterior distributions encompass both positive and negative values to a greater extent.

## 5. Discussion

## 5.1 Implications for theory

The results presented in the previous section suggest that considering environmentally-friendly certifications as a form of governance can help to disentangle their role as organisational tools along the supply chain.

From a TCE point of view, our results seem to support both of the hypotheses proposed in Section 2.1. Firstly, there is evidence that the VIVA certification is envisioned as a tool to curb internal risks. In this respect, wine companies seem to believe in the certification's ability to improve transparency, routines and liability distribution, to the extent that these features increase the probability that a firm will adopt the certification. Clearly, this does not mean that there might not be other forms of vertical coordination. In fact, the risk is probably managed further through the different forms of transaction governance within the wine supply chain, which is characterised by hierarchical control mechanisms and the persistence of economic relationships amongst supply chain agents over time (Montaigne and Coelho, 2012). Next, the adoption of an environmentally-friendly certification can be also framed within the TCE conceptual framework in terms of external risk. This time, however, the exposure to exogenous

## *Volume 27 · Number 7 · 2022 · 33–48*

threats discourages the implementation of the VIVA certification, as theory would recommend. Indeed, the procedures embedded in the VIVA system appear to negatively impact the probability that a firm will engage in an environmental certification when the external environment is more uncertain. This argument is in line with Stranieri et al. (2017), Wever et al. (2012) and the body of literature highlighting the importance of flexibility in a firm's vertical relationships (Olson and Wu, 2017; Shirmohammadi et al., 2020; Tang and Tomlin, 2008). Other studies have highlighted the importance of risks related to the external environment for the adoption of certifications. However, they have been interpreted only as control variables rather than following a specific theoretical framework. For example, Demirel et al. (2018) used customer requests as a dummy variable to explain the adoption of environmental management systems amongst UK firms, while Graafland and Gerlagh (2019) used price competition amongst the control variables to study firms' environmental performance.

Switching focus to the firm level, the first hypothesis based on the RBV theory concerned the role of the environmental certification when integrated as a resource possessing VRIN characteristics. Our results demonstrate that these characteristics do not play a clear role in predicting the probability that a firm will adopt the VIVA certification. Most likely, even in those firms with higher scores, these characteristics are not sufficiently relevant to induce any detectable effect. This reinforces the argument that environmentally-friendly certifications, including the one, studied here, aim at standardising practices and these practices are generic and imitable across firms (Treacy *et al.*, 2019). Consequently, they are not perceived as an asset that alone could produce a competitive advantage.

As hypothesis RBV-H4 suggests, the certification could leverage potential knowledge and be perceived as a promoter of operational capabilities. That is, the procedures required by the certification could enhance a firm's capabilities, thereby producing efficiencies at different levels of the production process, including input procurement, labour productivity and production costs (Delmas and Pekovic, 2013; Lo et al., 2014; Prajogo et al., 2014). However, our results do not appear to support this prediction, probably because managers were not interested in pursuing operational efficiency through the certification. As this study is not focussed on measuring performance, this does not mean that the certification cannot produce operational efficiencies - as found, for example, by Treacy et al. (2019) in the case of ISO14001 certification, by Lo et al. (2014) for the OHSAS 18001 certification, or, at a more general level, by Annunziata (2018) for proactive environmental practices in the Italian wine industry.

Finally, the adoption of certification may leverage dynamic capabilities that could promote a sustained competitive advantage over non-adopters. Hypotheses *RBV-H5* and *RBV-H6* investigate two of these capabilities, namely, those related to supply chain coordination skills and those related to the propensity for innovation. The evidence clearly supports the prediction in hypothesis *RBV-H5*. That is, managers are more likely to implement the certification when they expect that its adoption will improve their knowledge of the supply chain or prompt better and new collaborations with suppliers. As suggested by Hofmann *et al.* (2012), prior collaboration capabilities might play a role in

facilitating the adoption of certification or, more in general, of environmentally sustainable practice. However, our results say something different: managers expect the certification to create the basis to further develop collaborations along the supply chain. This concept is intrinsically dynamic and refers to the possibility of developing knowledge, that ex-ante might only be potential (Ghozzi et al., 2016), with a certification that leverages dynamic capabilities as defined by Wang and Ahmed (2007). These results are also consistent with firms pursuing a sustainable development strategy. As defined by Hart (1995, p. 1007) in his "Natural-RBV of the Firm", this strategy "facilitates and accelerates capability development" in environmentally-friendly activities and "vice versa". In fact, green capabilities can also offer strategic support: according to Hart (1995), collaboration with other actors in the supply chain and more in general with stakeholders can bring substantial technological change. On a different ground, results do not support RBV-H6 as there is no clear evidence of an association between the probability that a firm will adopt the VIVA certification and managerial expectations concerning innovation capacity. In other words, when the certification is envisioned as a tool fostering innovation, this has a very unclear impact on the choice to undergo certification. This result contrasts the empirical evidence provided by Prajogo et al. (2014) for green management systems.

To conclude, although from the results regarding *RBV-H4* the certification might not be considered a valuable, rare, inimitable or non-substitutable resource, it can interact with the capabilities of the firm, expressing complementarities that stimulate the formation of dynamic capabilities. These capabilities can be considered valuable from a managerial perspective, as they can lead to competitive advantage. This result supports the findings of Gavronski *et al.* (2011), which highlight how low-order standardised procedures from a certification like ISO14001 can be combined uniquely with the complex and higher-order capabilities of a firm, leading to better green supply management.

## 5.2 Implications for practice

From a TCE perspective, the results highlight how the decision to adopt the certification can be justified by the expectations of managers around more efficient management of vertical relationships and a reduction in behavioural uncertainty, while at the same time, in case of high environmental uncertainty, firms prefer to remain flexible and avoid the implementation of new standardised procedures. Using Santos and Eisenhardt's (2005, p. 498) terminology, in "high-velocity" environments, adaptability becomes central and boundary decisions become "path-breaking". Therefore, promoting the adoption of an environmental standard would require addressing the problem with firm flexibility. This could be achieved by shifting the focus, whereby the standardised procedures and consequent supply chain coordination move from an efficiency perspective to more strategic issues- and competence-based considerations.

Following the logic of competitive advantage, adopters expect to develop better capabilities through a certification that improves the management of the supply chain. However, the certification does not need to have VRIN characteristics. Competitiveness is perceived to emerge even from the standardised and replicable procedures included in a public or collective certification. The key is the complementarity of

## *Volume 27 · Number 7 · 2022 · 33–48*

certification with internal resources that can exploit potential knowledge developments. Under market dynamism, the capabilities related to supply chain coordination are not only important as an antecedent for the adoption of environmental initiatives, as Annunziata et al. (2018) find for the wine industry, but are expected to directly feed dynamic capabilities exploiting their potential in a virtuous cycle. Therefore, the development and promotion of certification such as VIVA can represent a useful tool to promote a long-run environmental strategy within a supply chain. If the expectations of early adopters are confirmed, this perception could spread to late adopters and improve the overall management of the supply chain and its environmental impact. In the case of the wine industry, this could be an important message for industry associations or focal firms supporting these initiatives within their supply chain. It implies delivering a message where the success of certification is not simply related to its marketing and differentiation content, but to its ability to foster the development of procedures that can find synergies with firm resources and exploit the organisational capability potential within the sector.

On similar grounds, stakeholders in the wine industry should also consider the lack of a significant effect of expectations concerning operational efficiency as an interesting signal. As highlighted by Pomarici et al. (2015), winemakers have invested considerably in recent years in production efficiency as a response to sustainability challenges. Improvements in wine sustainability are expected as a continuous process in winemaking, while grape production is expected to improve substantially because of new varieties in the pipeline (Pomarici and Vecchio, 2019). This might induce a general perception that a plateau will be reached in terms of operational efficiency, at least for those benefits that certification could one allow foreseeing. This would also reinforce the standpoint of winemakers who culturally focus on the operational dimension of a certification, considering it a "cost" rather than an opportunity (Pomarici et al., 2015). This suggests that, if an efficiency gain is the objective of certification, much more effort is needed to provide convincing evidence to supply chain agents regarding the operational benefits that certification could bring. This point is also reinforced when looking at the effect of the control variables included in the model. In fact, implementation costs have a negative impact on adoption, which highlights the care that entrepreneurs take regarding this aspect in their decision about whether to adopt.

Finally, the results presented here do not show significant evidence of an effect of winery size or for the presence of other environmentally-friendly certifications. Therefore, the potential targets for VIVA certification are independent of these characteristics. Interestingly, supply to a large-scale retailer strongly predicts the adoption of the certification. This result is in line with several findings in the literature (Chen *et al.*, 2015; Heyder *et al.*, 2012; Stranieri *et al.*, 2016) that empirically show how requests from retailers significantly affect a firm's decision to adopt a voluntary certification.

# 6. Concluding remarks

Environmentally-friendly certifications are gaining growing importance within food supply chains at the international level, even in the presence of heterogeneous formats that differ

*Volume 27 · Number 7 · 2022 · 33–48* 

substantially in terms of standards and amount of information labelled. In the wine sector, for instance, one can distinguish between voluntary public standards, i.e. organic certification at the European level and other national public standards, like the VIVA certification in Italy.

Although recent literature has explored the drivers leading firms to adopt such certifications, it has not closely examined the strategic motivations associated with their adoption. This paper investigated the VIVA certification, considering it an alternative form of supply chain governance. The aim was to investigate the drivers affecting the adoption of this certification by assessing managerial perceptions related to transactionrelated characteristics and firm-internal resources and capabilities. Specifically, our research question was explored within the conceptual boundaries of TCE and RBV. A Hierarchal Bayesian Model was applied to survey data based on a structured questionnaire submitted to all wine producers in charge of the decision about whether or not to adopt the certification.

Overall, our results provide some interesting insights when comparing TCE and RBV postulates. The certification does seem to work as a coordination tool to curb behavioural uncertainty amongst supply chain actors. However, results also suggest that in cases of perceived high external risks, managers prioritise structural flexibility over vertical coordination mechanisms (such as environmental certifications). Extending the analysis to RBV, considerations linked to operational capabilities do not seem to be relevant drivers of adoption. However, the supply chain coordination activities implied by the certification are perceived as important from a strategic perspective. There is a higher probability that a firm will adopt a certification when managers perceive that their dynamic supply chain management capabilities can be leveraged by the certification.

The study is, however, not free of limitations. Firstly, our results are highly sector-specific, so extrapolation hinges on the structural characteristics of the wine sector and the features of the increasing demand for quality attributes of wine products. Moreover, the study does not model the possible interactions amongst different transaction risks and resource- and capability-related drivers in affecting the decision of whether to adopt the certification. The simultaneous consideration of such different strategic variables could play an important role in better conceptualising firms' behaviour towards environmentally-friendly certifications. From a methodological perspective, some results might be skewed by the limited number of items per latent attribute in the RSM component of the HBM.

Future work could improve the external validity of this study by investigating the consistency of our discussion across other food and non-food sectors. In this respect, it could be also beneficial to test these results against other environmentallyfriendly certifications. The further empirical investigation should also address the interaction effect of different supply chains and firm organisational drivers to uncover multiple and possibly heterogeneous strategies for the adoption of voluntary certifications. Future surveys could also concentrate on longitudinal data to better represent and explain possible path dependencies in the strategic drivers behind the adoption of environmentally-friendly certifications.

## Notes

- 1 The other certifications, all promoted by private institutions, are Gea Vite/Ita.ca, Vino Libero, Equalitas and Tergeo. The first two also issue a label.
- 2 More information can be retrieved from the certification website: www.viticolturasostenibile.org/
- 3 For example, consider statements 1 through 3 in Table 1: as indicated in the second column, we specifically designed these statements to quantify the degree to which firms are exposed to internal risks. Model (1) accommodates internal risk through the  $\theta_j$  coefficient, which represents the firm-specific component of the RSM.
- 4 Following Table 1, we have: presence and severity of external risk and/or external risk, as well as attitude towards the potential benefits in terms of improved VRIN characteristics, efficiency, dynamic capabilities related to the innovation process and dynamic capabilities related to the supply chain.
- 5 MVN stands for Multivariate Normal Distribution, while the suffix "half-" indicates a zero-truncated distribution. Also, notice that we parametrise normal distributions in terms of their mean and standard deviation.
- 6 Replacing the half Normal(0,3) prior with half Cauchy(0,2) or Exponential(0.1) distributions has no impact on the model's estimates.
- 7 A brief sensitivity analysis (available upon request) shows that our results are essentially invariant to higher values of  $\sigma_{\delta}$  and  $\sigma_{\theta}$ , provided a proper rescaling of  $\theta$  before it enters the likelihood function. Common options for estimating RSMs in standard psychrometric analysis involve setting  $\sigma_{\theta} = 1$ ,  $\sigma_{\theta} = 3$  or occasionally  $\sigma_{\theta} \sim \text{Half} - \text{Normal}(0, 3)$ .
- 8 Given this requirement, we scale all the regressors in **z**, dummy variables included. Specifically, we transform binary predictors following the procedure suggested in Gelman *et al.* (2008).
- 9 We briefly discuss such methods in a supporting material available upon request to the authors, where we also analyse their performance in terms of the reliability and model fit of the estimates.

# References

- Amit, R. and Schoemaker, P.J.H. (1993), "Strategic assets and organizational rent", *Strategic Management Journal*, Vol. 14 No. 1, pp. 33-46, doi: 10.1002/smj.4250140105.
- Andersen, E.B. (1997), "The rating scale model", in Handbook of Modern Item Response Theory, Springer, New York, NY, pp. 67-84.
- Andrich, D. (1978), "A rating formulation for ordered response categories", *Psychometrika*, Vol. 43 No. 4, pp. 561-573.
- Annunziata, E., Pucci, T., Frey, M. and Zanni, L. (2018), "The role of organizational capabilities in attaining corporate sustainability practices and economic performance: evidence from Italian wine industry", *Journal of Cleaner Production*, Vol. 171, pp. 1300-1311, doi: 10.1016/j. jclepro.2017.10.035.

- Argyres, N.S. and Zenger, T.R. (2012), "Capabilities, transaction costs, and firm boundaries", *Organization Science* (*Providence, R.I.*), Vol. 23 No. 6, pp. 1643-1657, doi: 10.1287/orsc.1110.0736.
- Barney, J.B. (1986), "Strategic factor markets: expectations, luck, and business strategy", *Management Science*, Vol. 32 No. 10, pp. 1231-1241, doi: 10.1287/mnsc.32.10.1231.
- Barney, J.B. (2001), "Resource-based theories of competitive advantage: a ten-year retrospective on the resource-based view", *Journal of Management*, Vol. 27 No. 6, pp. 643-650, doi: 10.1016/S0149-2063(01)00115-5.
- Beske, P., Land, A. and Seuring, S. (2014), "Sustainable supply chain management practices and dynamic capabilities in the food industry: a critical analysis of the literature", *International Journal of Production Economics*, Vol. 152, pp. 131-143.
- Betancourt, M. (2017), "A conceptual introduction to hamiltonian monte carlo", arXiv preprint arXiv:1701.02434.
- Bonroy, O. and Constantatos, C. (2015), "On the economics of labels: how their introductionaffects the functioning of markets and the welfare of all participants", *American Journal* of *Agricultural Economics*, Vol. 97 No. 1, pp. 239-259.
- Borra, D. Dal Vecchio, A. Lazzari, A.M. Massaglia, S. and Viberti, A. (2016), "La percezione del concetto di sostenibilità nel settore vitivinicolo. FrancoAngeli".
- Bürkner, P.C. (2019), "Bayesian item response modeling in R with BRMS and STAN", arXiv preprint arXiv:1905.09501.
- Cassiman, B. and Veugelers, R. (2006), "In search of complementarity in innovation strategy: internal R&D and external knowledge acquisition", *Management Science*, Vol. 52 No. 1, pp. 68-82, doi: 10.1287/mnsc.1050.0470.
- Chen, E., Flint, S., Perry, P., Perry, M. and Lau, R. (2015), "Implementation of non-regulatory food safety management schemes in New Zealand: a survey of the food and beverage industry", *Food Control*, Vol. 47, pp. 569-576.
- Chkanikova, O. and Lehner, M. (2015), "Private eco-brands and green market development: towards new forms of sustainability governance in the food retailing", *Journal of Cleaner Production*, Vol. 107, pp. 74-84.
- Collis, D.J. and Montgomery, C.A. (1995), "Competing on resources: strategy in the 1990s", *Harvard Business Review*, Vol. 73, p. 118.
- Darnall, N. and Edwards, D. (2006), "Predicting the cost of environmental management system adoption: the role of capabilities, resources and ownership structure", *Strategic Management Journal*, Vol. 27 No. 4, pp. 301-320, doi: 10.1002/ smj.518.
- Das, T.K. and Teng, B.S. (2001), "Trust, control, and risk in strategic alliances: an integrated framework", *Organization Studies*, Vol. 22 No. 2, pp. 251-283.
- De Steur, H., Temmerman, H., Gellynck, X. and Canavari, M. (2020), "Drivers, adoption, and evaluation of sustainability practices in Italian wine SMEs", *Business Strategy and the Environment*, Vol. 29 No. 2, pp. 744-762.
- Delmas, M.A. and Pekovic, S. (2013), "Environmental standards and labor productivity: understanding the mechanisms that sustain sustainability", *Journal of Organizational Behavior*, Vol. 34 No. 2, pp. 230-252, doi: 10.1002/job.1827.
- Demirel, P., Iatridis, K. and Kesidou, E. (2018), "The impact of regulatory complexity upon self-regulation: evidence from the adoption and certification of environmental management

*Volume 27 · Number 7 · 2022 · 33–48* 

systems", *Journal of Environmental Management*, Vol. 207, pp. 80-91.

- Ding, H., Fu, Y., Zheng, L. and Yan, Z. (2019), "Determinants of the competitive advantage of dairy supply chains: evidence from the Chinese dairy industry", *International Journal of Production Economics*, Vol. 209, pp. 360-373.
- Dodds, R., Graci, S., Ko, S. and Walker, L. (2013), "What drives environmental sustainability in the New Zealand wine industry?", *International Journal of Wine Business Research.*, Vol. 25 No. 3.
- Engelhard, G. (2014), "Item response theory models for rating scale data", Wiley StatsRef: Statistics Reference, available at: https://doi.org/10.1002/9781118445112.stat06398
- Farber, J.M., Kozak, G.K. and Duquette, S. (2011), "Changing regulation: Canada's new thinking on listeria", *Food Control*, Vol. 22 No. 9, pp. 1506-1509.
- Fernández-Olmos, M., Rosell-Martínez, J. and Espitia-Escuer, M.A. (2009), "Vertical integration in the wine industry: a transaction costs analysis on the Rioja DOCa", *Agribusiness*, Vol. 25 No. 2, pp. 231-250.
- Forum per la sostenibilità del vino (FSV) (2014), "Primo rapporto sulla sostenibilità del vino., ottobreOctober 2014", available at: www.vinosostenibile.org/wp-content/uploads/2014/10/Primo-Rapporto-Sostenibilita-del-Vino-Ottobre-2014. pdf (last accessed\_November\_2020).
- Furr, D.C. (2017), "Edstan: Stan models for item response theory. R package version 1.0.6", available at: https://CRAN. R-project.org/package=edstan
- Gavronski, I., Klassen, R.D., Vachon, S. and Nascimento, L.F. M.D. (2011), "A resource-based view of green supply management", *Transportation Research Part E: Logistics and Transportation Review*, Vol. 47 No. 6, pp. 872-885, doi: 10.1016/j.tre.2011.05.018.
- Gelman, A., Simpson, D. and Betancourt, M. (2017), "The prior can often only be understood in the context of the likelihood", *Entropy*, Vol. 19 No. 10, p. 555.
- Gelman, A., Jakulin, A., Pittau, M.G. and Su, Y.S. (2008), "A weakly informative default prior distribution for logistic and other regression models", *The Annals of Applied Statistics*, Vol. 2 No. 4, pp. 1360-1133.
- Gelman, A., Stern, H.S., Carlin, J.B., Dunson, D.B., Vehtari, A. and Rubin, D.B. (2013), *Bayesian Data Analysis*, Chapman and Hall/CRC.
- Geyer, C. (2011), "Introduction to Markov chain Monte Carlo", Handbook of Markov Chain Monte Carlo, Vol. 20116022, p. 45.
- Ghosh, J., Li, Y. and Mitra, R. (2018), "On the use of Cauchy prior distributions for Bayesian logistic regression", *Bayesian Analysis*, Vol. 13 No. 2, pp. 359-383.
- Ghozzi, H., Soregaroli, C., Boccaletti, S. and Sauvée, L. (2016), "Impacts of non-GMO standards on poultry supply chain governance: transaction cost approach vs resourcebased view", *Supply Chain Management: An International Journal*, Vol. 21 No. 6, pp. 743-758, doi: 10.1108/SCM-03-2016-0089.
- Gilinsky, A., Newton, S.K. and Vega, R.F. (2016), "Sustainability in the global wine industry: concepts and cases", *Agriculture and Agricultural Science Procedia*, Vol. 8 No. 1, pp. 37-49.
- Graafland, J. and Gerlagh, R. (2019), "Economic freedom, internal motivation, and corporate environmental responsibility

of SMEs", Environmental and Resource Economics, Vol. 74 No. 3, pp. 1101-1123.

Grant, R.M. (1991), "The resource-based theory of competitive advantage: implications for strategy formulation", *California Management Review*, Vol. 33 No. 3, pp. 114-135, doi: 10.2307/41166664.

Grimm, J.H., Hofstetter, J.S. and Sarkis, J. (2014), "Critical factors for Sub-supplier management: a sustainable food supply chains perspective", *International Journal of Production Economics*, Vol. 152, pp. 159-173.

Gurcaylilar-Yenidogan, T. and Windsperger, J. (2014), "Interorganizational performance in the automotive supply networks: the role of environmental uncertainty, specific investments and formal contracts", *Procedia - Social and Behavioral Sciences*, Vol. 150, pp. 813-822.

Hammoudi, A., Grazia, C. and Hamza, O. (2015), "On the emergence of private standards: an industrial organization approach", In: Abdelhakim Hammoudi, Cristina Grazia, Yves Surry, Jean-Baptiste Traversac, Dir., Food Safety, Market Organization, Trade and Development, FRA: Editions Springer, (p. 77-109).

Hart, S.L. (1995), "A natural-resource-based view of the firm", *Academy of Management Review*, Vol. 20 No. 4, pp. 986-1014, doi: 10.2307/258963.

Hartmann, M., Frohberg, K. and Fischer, C. (2010), "Building sustainable relationships in agri-food chains: challenges from farm to retail", in Hartmann, M. and Fischer, C.(Eds), agri-Food Chain Relationships, CAB International, pp. 25-44.

Hau, H. (2006), "The role of transaction costs for financial volatility: evidence from the Paris bourse", *Journal of the European Economic Association*, Vol. 4 No. 4, pp. 862-890.

Hauschildt, V. and Schulze-Ehlers, B. (2014), "An empirical investigation into the adoption of green procurement practices in the German food service industry", *International Food and Agribusiness Management Review*, Vol. 17 No. 1030-2016-83029, pp. 1-32.

Heyder, M., Theuvsen, L. and Hollmann-Hespos, T. (2012), "Investments in tracking and tracing systems in the food industry: a PLS analysis", *Food Policy*, Vol. 37 No. 1, pp. 102-113.

Hoff, P.D. (2009), A First Course in Bayesian Statistical Methods, (Vol. 580). New York, NY: Springer.

Hofmann, K.H., Theyel, G. and Wood, C.H. (2012), "Identifying firm capabilities as drivers of environmental management and sustainability practices – evidence from small and medium-sized manufacturers", *Business Strategy and the Environment*, Vol. 21 No. 8, pp. 530-545, doi: 10.1002/bse.739.

Huchet-Bourdon, M. (2011), "Agricultural commodity PriceVolatility: an overview", OECD Food, Agriculture andFisheries Working Papers, No. 52, OECD Publishing.

Italian National Institute of Statistics (ISTAT) (2019), "Trend of the agricultural economy, year 2018", available at: www. istat.it/en/archivio/230673.

Jiménez-Jiménez, D. and Sanz-Valle, R. (2011), "Innovation, organizational learning, and performance", *Journal of Business Research*, Vol. 64 No. 4, pp. 408-417, doi: 10.1016/j. jbusres.2010.09.010.

Johnsen, T.E., Miemczyk, J. and Howard, M. (2017), "A systematic literature review of sustainable purchasing and supply research: theoretical perspectives and opportunities

for IMP-based research", *Industrial Marketing Management*, Vol. 61, pp. 130-143.

- Jose, A. and Shanmugam, P. (2019), "Supply chain issues in SME food sector: a systematic review", *Journal of Advances in Management Research.*, Vol. 17 No. 1.
- Jourjon, F., Chou, H.C., Gezart, A., Kadison, A.E., Martinat, L., Pomarici, E. and Vecchio, R. (2016), "Wineries evaluation of costs and benefits of sustainability certification program: the case of terra Vitis in France", *Recent Patents on Food, Nutrition & Agriculture*, Vol. 8 No. 2, pp. 138-147.
- Kaplan, R.S. and Mikes, A. (2012), "Managing risks: a new framework", *Harvard Business Review*, Vol. 90 No. 6, pp. 48-60.
- Kruschke, J. (2014), Doing Bayesian Data Analysis: A Tutorial with R, JAGS, and Stan, Academic Press.
- Kvalvik, I., Dalmannsdottir, S., Dannevig, H., Hovelsrud, G., Rønning, L. and Uleberg, E. (2011), "Climate change vulnerability and adaptive capacity in the agricultural sector in Northern Norway", Acta Agriculturae Scandinavica, Section B-Soil & Plant Science, Vol. 61 No. sup1, pp. 27-37.
- Lee, C.H., Wahid, N.A. and Goh, Y.N. (2013), "Perceived drivers of green practices adoption: a conceptual framework", *Journal of Applied Business Research (JABR)*, Vol. 29 No. 2, pp. 351-360.
- Lemoine, N.P. (2019), "Moving beyond noninformative priors: why and how to choose weakly informative priors in Bayesian analyses", *Oikos*, Vol. 128 No. 7, pp. 912-928.
- Lo, C.K.Y., Yeung, A.C.L. and Cheng, T.C.E. (2009), "ISO 9000 and supply chain efficiency: empirical evidence on inventory and account receivable days", *International Journal* of Production Economics, Vol. 118 No. 2, pp. 367-374, doi: 10.1016/j.ijpe.2008.11.010.
- Lo, C.K.Y., Pagell, M., Fan, D., Wiengarten, F. and Yeung, A. C.L. (2014), "OHSAS 18001 certification and operating performance: the role of complexity and coupling", *Journal of Operations Management*, Vol. 32 No. 5, pp. 268-280, doi: 10.1016/j.jom.2014.04.004.
- Luo, Y. and Jiao, H. (2018), "Using the stan program for Bayesian item response theory", *Educational and Psychological Measurement*, Vol. 78 No. 3, pp. 384-408.
- McElreath, R. (2020), Statistical Rethinking: A Bayesian Course with Examples in R and Stan, CRC press.
- Mangla, S.K., Luthra, S., Rich, N., Kumar, D., Rana, N.P. and Dwivedi, Y.K. (2018), "Enablers to implement sustainable initiatives in agri-food supply chains", *International Journal of Production Economics*, Vol. 203, pp. 379-393.
- Marette, S. (2016), "Label proliferation and consumers' confusion. The weak impact of sustain-ability labels on the value of chocolate", *SUSDIET Newsletter*, Vol. 4, p. 5.
- Ménard (2017), Finding Our Way in the Jungle: insights from Organization Theory in Martino G., Karantininis K., Pascucci S., Dries L. and Codron J.M. It's a Jungle out There – the Strange Animals of Economic Organization in Agri-Food Value Chains, Wageningen academic Publishers, pp. 27-50.
- Montaigne, E. and Coelho, A. (2012), "Structure of the producing side of the wine industry: firm typologies, networks of firms and clusters", *Wine Economics and Policy*, Vol. 1 No. 1, pp. 41-53, doi: 10.1016/j.wep.2012.12.002.
- Nagano, H. (2020), "The growth of knowledge through the resource-based view", *Management Decision*, Vol. 58 No. 1, pp. 98-111, doi: 10.1108/md-11-2016-0798.

- Naidoo, M. and Gasparatos, A. (2018), "Corporate environmental sustainability in the retail sector: drivers, strategies and performance measurement", *Journal of Cleaner Production*, Vol. 203, pp. 125-142.
- Notarnicola, B., Hayashi, K., Curran, M.A. and Huisingh, D. (2012), "Progress in working towards a more sustainable agri-food industry", *Journal of Cleaner Production*, Vol. 28, pp. 1-8.
- Olmos, M.F. (2010), "The performance implications of 'grow or buy' decisions in the wine industry", *Food Policy*, Vol. 35 No. 3, pp. 256-264.
- Olson, D.L. and Wu, D.D. (2017), "Sustainability and enterprise risk management", In *Enterprise Risk Management Models*, (pp. 193-204). Springer, Berlin, Heidelberg.
- Passuello, F., Boccaletti, S. and Soregaroli, C. (2015), "Governance implications of non-GMO private standards on poultry meat value chains", *British Food Journal*, Vol. 117 No. 10, pp. 2564-2581.
- Perrow, C. (1986), *Complex Organizations*, (3rd ed.), New York, NY: Random House.
- Pomarici, E. and Vecchio, R. (2019), "Will sustainability shape the future wine market?", *Wine Economics and Policy*, Vol. 8 No. 1, pp. 1-4, doi: 10.1016/j.wep.2019.05.001.
- Pomarici, E., Vecchio, R. and Mariani, A. (2015), "Wineries' perception of sustainability costs and benefits: an exploratory study in California", *Sustainability*, Vol. 7 No. 12, pp. 16164-16174.
- Prajogo, D., Tang, A.K.Y. and Lai, K. (2014), "The diffusion of environmental management system and its effect on environmental management practices", *International Journal of Operations & Production Management*, Vol. 34 No. 5, pp. 565-585, available at: www.econis.eu/PPNSET?PPN=79100841X.
- Pullman, M.E., Maloni, M.J. and Carter, C.R. (2009), "Food for thought: social versus environmental sustainability practices and performance outcomes", *Journal of Supply Chain Management*, Vol. 45 No. 4, pp. 38-54.
- Rasch, G. (1961), "On general laws and the meaning of measurement in psychology", In *Proceedings of the* fourth Berkeley symposium on mathematical statistics and probability, Vol. 4, pp. 321-333.
- Rasch, G. (1977), "On specific objectivity. An attempt at formalizing the request for generality and validity of scientific statements in symposium on scientific objectivity, Vedbaek, Mau 14-16, 1976", Danish Year-Book of Philosophy Kobenhavn, Vol. 14, pp. 58-94.
- Romme, A.G.L. (1990), "Vertical integration as organizational strategy formation", Organization Studies, Vol. 11 No. 2, pp. 239-260.
- Santos, F.M. and Eisenhardt, K.M. (2005), "Organizational boundaries and theories of organization", Organization Science (Providence, R.I.), Vol. 16 No. 5, pp. 491-508, doi: 10.1287/orsc.1050.0152.
- Schulze, B., Spiller, A. and Theuvsen, L. (2007), "A broader view on vertical coordination: lessons from German pork production", *Journal on Chain and Network Science*, Vol. 7 No. 1, pp. 35-53.
- Shirmohammadi, A., Vafae, F., Namamian, F. and Taban, M. (2020), "Developing a business

sustainability model in the supply chain using the Meta-Synthesis approach", *Journal of Business Management*, Vol. 12 No. 3, pp. 627-651.

- Stan Development Team (2018), "Stan modeling language users guide and reference manual, version 2.18.0", available at: http://mc-stan.org
- Stranieri, S., Orsi, L. and Banterle, A. (2017), "Traceability and risks: an extended transaction cost perspective", *Supply chain management An International Journal*, Vol. 22 No. 2, pp. 145-159.
- Stranieri, S., Orsi, L., Banterle, A. and Ricci, E.C. (2019), "Sustainable development and supply chain coordination: the impact of corporate social responsibility rules in the European union food industry", *Corporate Social Responsibility and Environmental Management*, Vol. 26 No. 2, pp. 481-491.
- Stranieri, S., Cavaliere, A. and Banterle, A. (2016), "Voluntary traceability standards and the role of economic incentives", *British Food Journal*, Vol. 118 No. 5, pp. 1025-1040, doi: 10.1108/BFJ-04-2015-0151.
- Sydow, J., Müller-Seitz, G. and Provan, K.G. (2013), "Managing uncertainty in alliances and networks – from governance to practice", *Managing Knowledge in Strategic Alliances*, pp. 1-43.
- Tang, C. and Tomlin, B. (2008), "The power of flexibility for mitigating supply chain risks", *International Journal of Production Economics*, Vol. 116 No. 1, pp. 12-27.
- Teece, D.J. (2009), Dynamic Capabilities and Strategic Management: Organizing for Innovation and Growth, Oxford University Press, Oxford.
- Treacy, R., Humphreys, P., McIvor, R. and Lo, C. (2019), "ISO14001 certification and operating performance: a practice-based view", *International Journal of Production Economics*, Vol. 208, pp. 319-328, doi: 10.1016/j. ijpe.2018.12.012.
- Vermeulen, W.J. and Ras, P.J. (2006), "The challenge of greening global product chains: meeting both ends", *Sustainable Development*, Vol. 14 No. 4, pp. 245-256.
- Wang, C.L. and Ahmed, P.K. (2007), "Dynamic capabilities: a review and research agenda", *International Journal of Management Reviews*, Vol. 9 No. 1, pp. 31-51, doi: 10.1111/ j.1468-2370.2007.00201.x.
- Wernerfelt, B. (1984), "A resource-based view of the firm", Strategic Management Journal, Vol. 5 No. 2, pp. 171-180, doi: 10.1002/smj.4250050207.
- Wernerfelt, B. (1989), "From critical resources to corporate strategy", *Journal of General Management*, Vol. 14 No. 3, p. 4, available at: https://search.proquest.com/docview/ 1311919261
- Wever, M., Wognum, P.M., Trienekens, J.H. and Omta, S.W. F. (2012), "Supply chain-wide consequences of transaction risks and their contractual solutions: towards an extended transaction cost economics framework", *Journal of Supply Chain Management*, Vol. 48 No. 1, pp. 73-91.
- Williamson, O.E. (1985), *The Economics of Capitalism*, Free Press, New York, NY.
- Williamson, O.E. (1991), "Comparative economic organization: the analysis of discrete structural alternatives", *Administrative Science Quarterly*, Vol. 36 No. 2, pp. 269-296.

Italian wine supply chain

Stefanella Stranieri et al.

*Volume 27 · Number 7 · 2022 · 33–48* 

- Yin, R.K. (2017), Case Study Research and Applications: Design and Methods, Sage publications.
- Yokessa, M. and Marette, S. (2019), "A review of eco-labels and their economic impact", *International Review of Environmental and Resource Economics*, Vol. 13 Nos 1/2, pp. 119-163.

# **Further reading**

Bayarri, M.J. and Castellanos, M.E. (2007), "Bayesian checking of the second levels of hierarchical models", *Statistical Science*, Vol. 22 No. 3, pp. 322-343.

- Betancourt, M. and Girolami, M. (2015), "Hamiltonian Monte Carlo for hierarchical models", *Current Trends in Bayesian Methodology with Applications*, Vol. 79, p. 30.
- Daniel, M. and Alastair, R. (2018), "Comparing wine sustainability certifications around the world: history, status and opportunity", *Journal of Wine Research*, Vol. 29 No. 1, pp. 1-25.
- MacEachern, S.N. and Berliner, L.M. (1994), "Subsampling the Gibbs sampler", *The American Statistician*, Vol. 48 No. 3, pp. 188-190.

# **Corresponding author**

**Claudio Soregaroli** can be contacted at: claudio.soregaroli@ unicatt.it

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com